Author's Response To Reviewer Comments

Clo<u>s</u>e

Dear Editor,

as requested in point (1), a reference to our GigaDB dataset has been included in the bibliography with the number 33.

We also made the necessary corrections requested in point (2) and uploaded a revised manuscript.

Best regards,

Tiziana Castrignanò

Clo<u>s</u>e